Calculus I Russell Brown The last handout December 4, 2000 Our final is at 8:30-10:30 pm on Monday, 11 December, 2000. The final will be in CB110. An informal problem session will take place at 9:30-11am on Monday 11 December in the Math House, on Columbia Ave, across from W.T. Young library. I have already said enough. The point of this session is to have one last chance to ask questions and work problems.

• The final will concentrate on the material since the third test. There is approximately 50% new material and 50% material from tests 1-3.

• You will not be asked to state definitions and state or prove theorems from the material covered by the first three exams. This does not mean that you should forget the product rule or forget the definition of continuity.

• The final from last year is on the web and in the library.

• Review problems from recitation and homework. Below is a selection of problems from the sample test problems (not the concept tests) at the end of each chapter.

– §2.11 #33 – §3.11 3ab, 5, 7, 23, 45, 46 – §4.8 #1, 5, 22 – §5.9 #1–11, 13, 16, 40 – §6.7 #1-5, 11, 18–20